Author's response to reviews

Title: Participation levels of physical activity programs for community-dwelling older adults: a systematic review

Authors:
Marielle van der Deijl (mariellevanderdeijl@hotmail.com)
Astrid Etman (a.etman@erasusmc.nl)
Carlijn BM Kamphuis (c.kamphuis@erasusmc.nl)
Frank J van Lenthe (f.vanlenthe@erasusmc.nl)

Version: 2 Date: 23 July 2014

Author's response to reviews:

Dear editor,

We apologize the first submission of the manuscript did not adhere to the PRISMA guidelines. As requested, we included the PRISMA checklist and PRISMA flowchart.

Hopefully the manuscript will be considered for publication.

With kind regards,

Astrid Etman